\documentclass[11pt]{article}
\usepackage{graphicx}
\usepackage{amssymb}
\usepackage{fancyhdr}
\usepackage{tikz}
\usepackage{scrextend}
\usepackage{enumerate}
\pagestyle{empty}



\textwidth = 6.5 in \textheight = 9.5 in \oddsidemargin = 0.0 in
\evensidemargin = 0.0 in \topmargin = -.25 in \headheight = 0.0 in
\headsep = 0.0 in
\parskip = 0.1in
\parindent = 0.0in

\newcommand{\contradiction}{{\hbox{%
    \setbox0=\hbox{$\mkern-3mu\times\mkern-3mu$}%
    \setbox1=\hbox to0pt{\hss$\times$\hss}%
    \copy0\raisebox{0.5\wd0}{\copy1}\raisebox{-0.5\wd0}{\box1}\box0
}}}

\newcommand\Z{\mathbb Z}
\newcommand\Q{\mathbb Q}
\newcommand\R{\mathbb R}
\newcommand\N{\mathbb N}
\newcommand\M{\mathbb M}
\newcommand\C{\mathbb C}
\newcommand\odd{\mathbb O}
\newcommand\E{\mathbb E}
\newcommand\vp{\varphi}
\newcommand*{\QED}{\hfill\ensuremath{\square}}%
 \newcommand{\tab}{\hspace*{2em}}
\newcommand{\Divides}{\mathbin{\big|}}





\title{FERPA Waiver: Written Homework}
\author{Math 23: Differential Equations\\ Fall 2021 \\ Anne Gelb and Yao Xiao}
\begin{document}

\maketitle


By my signature, I relinquish my FERPA rights in the following context: My written homework for Math 23, Fall 2021, may be returned en masse with others in the class via the homework boxes in Kemeny Hall.  I understand that my homework score may be visible to others. If  I  choose  not  to  relinquish  my  FERPA  rights,  I  understand  that  I  will  have  to present my student ID (for example at my instructor's office) in order to receive my exams and homework sets. 


\

\

Name (Print) : \hrulefill 


\

\


Signature: \hrulefill


\end{document}
